Author's response to reviews

Title: A multicenter prospective trial evaluating fetal bovine dermal graft (Xenform(R) Matrix) for pelvic reconstructive surgery

Authors:

Howard B. Goldstein (hgoldstein2009@gmail.com)
Joseph Maccarone (maccarone-joseph@cooperhealth.edu)
Martin J. Naughton (min@renogynecology.com)
Oscar A. Aguirre (oscar@milenstonemedicalresearch.com)
Rakesh C. Patel (rakeshpatel@wpuafl.com)

Version: 4 Date: 10 December 2010

Author's response to reviews:

10 December 2010
Melissa Norton
Editor-in-Chief
BMC Urology
Middlesex House
34-42 Cleveland Street
London, W1T 4LB United Kingdom
Re: Manuscript Number 1880235769397992
Dear Dr. Norton,
We are very pleased to hear of the acceptance of our manuscript entitled A Multicenter Prospective Trial Evaluating Fetal Bovine Dermal Graft (Xenform® Matrix) For Pelvic Reconstructive Surgery.
Per the request from the editors, we have made the required revisions to the manuscript proofs and are hereby uploading the revised proofs to the journal web site.
Sincerely,
Howard B. Goldstein, DO, MPH, FACOG
Corresponding Author
Director of Research and Education
Center for Urogynecology and Pelvic Surgery
4735 Ogletown-Stanton Road
MAP II, Suite 1208